

Microbiology Spectrum

Species-specific analysis of bacterial vaginosis (BV)-associated bacteria (BVAV)

John Osei Sekyere, Ayodeji Oyenihi, Jason Trama, and Martin Adelson

Corresponding Author(s): John Osei Sekyere, University of Pretoria

Review Timeline:	Submission Date:	November 16, 2022
	Editorial Decision:	April 12 2023

Editorial Decision: April 12, 2023
Revision Received: May 11, 2023
Accepted: June 4, 2023

Editor: Steven Frese

Reviewer(s): Disclosure of reviewer identity is with reference to reviewer comments included in decision letter(s). The following individuals involved in review of your submission have agreed to reveal their identity: Francescopaolo Antonucci (Reviewer #1)

Transaction Report:

(Note: With the exception of the correction of typographical or spelling errors that could be a source of ambiguity, letters and reports are not edited. The original formatting of letters and referee reports may not be reflected in this compilation.)

DOI: https://doi.org/10.1128/spectrum.04676-22

1st Editorial Decision April 12,

April 12, 2023

Dr. John Osei Sekyere University of Pretoria Department of Dermatology School of Medicine Faculty of Health Sciences Pretoria, Gauteng 0084 South Africa

Re: Spectrum04676-22 (Species-specific analysis of bacterial vaginosis (BV)-associated bacteria (BVAV))

Dear Dr. John Osei Sekyere:

I know this has taken some time, but thank you for your patience. I'd like to move forward with the reviews we do have below.

Thank you for submitting your manuscript to Microbiology Spectrum. When submitting the revised version of your paper, please provide (1) point-by-point responses to the issues raised by the reviewers as file type "Response to Reviewers," not in your cover letter, and (2) a PDF file that indicates the changes from the original submission (by highlighting or underlining the changes) as file type "Marked Up Manuscript - For Review Only". Please use this link to submit your revised manuscript - we strongly recommend that you submit your paper within the next 60 days or reach out to me. Detailed instructions on submitting your revised paper are below.

Link Not Available

Below you will find instructions from the Microbiology Spectrum editorial office and comments generated during the review.

ASM policy requires that data be available to the public upon online posting of the article, so please verify all links to sequence records, if present, and make sure that each number retrieves the full record of the data. If a new accession number is not linked or a link is broken, provide production staff with the correct URL for the record. If the accession numbers for new data are not publicly accessible before the expected online posting of the article, publication of your article may be delayed; please contact the ASM production staff immediately with the expected release date.

The ASM Journals program strives for constant improvement in our submission and publication process. Please tell us how we can improve your experience by taking this quick <u>Author Survey</u>.

Sincerely,

Steven Frese

Editor, Microbiology Spectrum

Journals Department American Society for Microbiology 1752 N St., NW Washington, DC 20036 E-mail: spectrum@asmusa.org

Reviewer comments:

Reviewer #1 (Comments for the Author):

Reviewer' Comments to Author:

The authors determined the actual species of Bacterial-Vaginosis associated Bacteria (BVAB) -1, -2 and -3 through phylogenetic analysis of their respective DNA sequences.

From this article it can be concluded that the species of BVAB-1 is Clostridials genomosp., BVAB-2 is Oscillospiraceae bacterium, and BVAB-3 is Mageeibacillus indolicus.

I due have some questions/suggestions to clarify few aspects and about the interpretation of the results:

QUESTION 1= How many sequences have you downloaded from GenBank?

QUESTION 2= Have you assessed the number of taxa among the used libraries?

QUESTION 3= Which were the characteristics of the patients from whom these strains were isolated? Were they all with a diagnosed BV? More discussion should be written about it in the "RESULTS AND DISCUSSION" section.

LINE 18-19= Please consider to readapt this sentence. Is BVAB-1 a O. bacterium or C. genomospecies? What about M. indolicus?

LINE 34= Please, provide reference.

LINE 48= Please, provide reference.

LINE 52= Please, provide reference.

LINE 68-76= Have you noticed any phylogenetic similarities of BVAB-1 with Shuttleworthia satelles? Were there any sequences of S. satelles that clustered with the BVAB-1 sequences? Considering resent results on literature, this should be clarified or at least mentioned.

LINE 74= Please, provide reference and add it also into the "REFERENCES" paragraph.

LINE 76= Please, check the correct reference and report it.

LINE 93= It would be worth to add a phylogenetic tree that includes not just BVAB-1, -2 and -3 but also some of the sequences that are located nearby this cluster.

LINE 94-96= Please, consider to clarify this sentence; are they same species or strains? Are they different species? Is BVAB-2 and -3 O. bacterium and M. indolicus, respectively?

FIGURE 1= I am assuming that the Figure 1 is the figure located at page 7. Please, consider to explain better this Figure description. Is the BVAB-1 the isolate UAB071? What about the purple clade?

FIGURE 2= I would suggest to colour the BVAB-3 strain with a different colour like you did for BVAB-2; this will help the identification of it among all the branches.

FIGURE S1= Are these sequences included in the phylogenetic tree of Figure 2 as well? I would recommend to include these sequences into the Figure 2 for better view of these clusters.

RESULTS AND DISCUSSION= Please, report in this section the results and their interpretations only; it is recommended not to mention further facts, such as the current methods used for the BVAB diagnosis. I would recommend to focus the attention more on the potential novel identified species, giving further explanation of the obtained results.

Staff Comments:

Preparing Revision Guidelines

To submit your modified manuscript, log onto the eJP submission site at https://spectrum.msubmit.net/cgi-bin/main.plex. Go to Author Tasks and click the appropriate manuscript title to begin the revision process. The information that you entered when you first submitted the paper will be displayed. Please update the information as necessary. Here are a few examples of required updates that authors must address:

- Point-by-point responses to the issues raised by the reviewers in a file named "Response to Reviewers," NOT IN YOUR COVER LETTER.
- Upload a compare copy of the manuscript (without figures) as a "Marked-Up Manuscript" file.
- Each figure must be uploaded as a separate file, and any multipanel figures must be assembled into one file.
- Manuscript: A .DOC version of the revised manuscript
- Figures: Editable, high-resolution, individual figure files are required at revision, TIFF or EPS files are preferred

For complete guidelines on revision requirements, please see the journal Submission and Review Process requirements at https://journals.asm.org/journal/Spectrum/submission-review-process. **Submissions of a paper that does not conform to Microbiology Spectrum guidelines will delay acceptance of your manuscript.** "

Please return the manuscript within 60 days; if you cannot complete the modification within this time period, please contact me. If you do not wish to modify the manuscript and prefer to submit it to another journal, please notify me of your decision immediately so that the manuscript may be formally withdrawn from consideration by Microbiology Spectrum.

If your manuscript is accepted for publication, you will be contacted separately about payment when the proofs are issued; please follow the instructions in that e-mail. Arrangements for payment must be made before your article is published. For a complete list of **Publication Fees**, including supplemental material costs, please visit our website.

Corresponding authors may join or renew ASM membership to obtain discounts on publication fees. Need to upgrade your membership level? Please contact Customer Service at Service@asmusa.org.

Thank you for submitting your paper to Microbiology Spectrum.

Reviewer' Comments to Author:

The authors determined the actual species of Bacterial-Vaginosis associated Bacteria (BVAB) -1, -2 and -3 through phylogenetic analysis of their respective DNA sequences. From this article it can be concluded that the species of BVAB-1 is Clostridials genomosp., BVAB-2 is Oscillospiraceae bacterium, and BVAB-3 is Mageeibacillus indolicus.

I due have some questions/suggestions to clarify few aspects and about the interpretation of the results:

QUESTION 1= How many sequences have you downloaded from GenBank?

QUESTION 2= Have you assessed the number of taxa among the used libreries?

QUESTION 3= Which were the characteristics of the patients from whom these strains were isolated? Were they all with a diagnosed BV? More discussion should be written about it in the "RESULTS AND DISCUSSION" section.

LINE 18-19= Please consider to readapt this sentence. Is BVAB-1 a O. bacterium or C. genomospecies? What about M. indolicus?

LINE 34= Please, provide reference.

LINE 48= Please, provide reference.

LINE 52= Please, provide reference.

LINE 68-76= Have you noticed any phylogenetic similarities of BVAB-1 with Shuttleworthia satelles? Were there any sequences of S. satelles that clustered with the BVAB-1 sequences? Considering resent results on literature, this should be clarified or at least mentioned.

LINE 74= Please, provide reference and add it also into the "REFERENCES" paragraph.

LINE 76= Please, check the correct reference and report it.

LINE 93= It would be worth to add a phylogenetic tree that includes not just BVAB-1, -2 and -3 but also some of the sequences that are located nearby this cluster.

LINE 94-96= Please, consider to clarify this sentence; are they same species or strains? Are they different species? Is BVAB-2 and -3 O. bacterium and M. indolicus, respectively?

FIGURE 1= I am assuming that the Figure 1 is the figure located at page 7. Please, consider to explain better this Figure description. Is the BVAB-1 the isolate UAB071? What about the purple clade?

FIGURE 2= I would suggest to colour the BVAB-3 strain with a different colour like you did for BVAB-2; this will help the identification of it among all the branches.

FIGURE S1= Are these sequences included in the phylogenetic tree of Figure 2 as well? I would recommend to include these sequences into the Figure 2 for better view of these clusters.

RESULTS AND DISCUSSION= Please, report in this section the results and their interpretations only; it is recommended to not mention further facts, such as the current methods used for the BVAB diagnosis. I would recommend to focus the attention more on the potential novel identified species, giving further explanation of the obtained results.

Reviewer comments	Response
QUESTION 1= How many sequences have you downloaded from GenBank?	Three 16S rRNA sequences, one each for BVAB-1, BVAB-2, and BVAB-3, were downloaded from GenBank for determining the taxa associated with them. See line 56
QUESTION 2= Have you assessed the number of taxa among the used libraries?	The taxa that aligned to the BVAB-1, BVAB-2, and BVAB-3 were mainly uncultured and undefined so it's impossible to assess the number of taxa. However, at least 100 sequences were closely aligned to the BLASTed sequences. For BVAB-1, the main taxa besides C. genomospo were Lachnospiraceae bacterium. See lines 69-71. For BVAB-2 & BVAB-3, the other taxa are detailed in lines 82-84, and 90-93.
QUESTION 3= Which were the characteristics of the patients from whom these strains were isolated? Were they all with a diagnosed BV? More discussion should be written about it in the "RESULTS AND DISCUSSION" section.	These strains were not isolated from patients, but they were identified from specimens obtained from patients with BV (bacterial vaginosis). Lines 59-60
LINE 18-19= Please consider to readapt this sentence. Is BVAB-1 a O. bacterium or C. genomospecies? What about M. indolicus?	Thanks. I modified these lines accordingly.
LINE 34= Please, provide reference.	Please note that line 34 already has an in-text citation: Fredericks et al. (2005)
LINE 48= Please, provide reference.	Provided in line 50
LINE 52= Please, provide reference.	Provided in line 53
LINE 68-76= Have you noticed any phylogenetic similarities of BVAB-1 with Shuttleworthia satelles? Were there any sequences of S. satelles that clustered with the BVAB-1 sequences? Considering resent results on literature, this should be clarified or at least mentioned.	Thanks for bringing this to our attention. Please see lines 101-110 for these details.
LINE 74= Please, provide reference and add it also into the "REFERENCES" paragraph.	An in-text citation of this is found in line 80. Also provided in the reference list.

LINE 76= Please, check the correct reference and report it.	Provided in line 82
LINE 93= It would be worth to add a phylogenetic tree that includes not just BVAB-1, -2 and -3 but also some of the sequences that are located nearby this cluster.	These are already shown in Figures 1, 2, and S1.
LINE 94-96= Please, consider to clarify this sentence; are they same species or strains? Are they different species? Is BVAB-2 and -3 O. bacterium and M. indolicus, respectively?	Thanks for drawing our attention to this. I added "respectively" to it to distinguish it. We used strains/species because some of them are strains while others are species. For instance, <i>M. indolicus</i> is species while O. bacterium CHIC02 is a strain. As these organisms are still being investigated, it's best to use strains and species to distinguish them as the same strains can even be later found to belong to different species.
FIGURE 1= I am assuming that the Figure 1 is the figure located at page 7. Please, consider to explain better this Figure description. Is the BVAB-1 the isolate UAB071? What about the purple clade?	Yes. Figure 1 is located at page 7. Thanks for your recommendation. We elaborated the figure legend/description in lines 161-165.
FIGURE 2= I would suggest to colour the BVAB-3 strain with a different colour like you did for BVAB-2; this will help the identification of it among all the branches.	Thanks. I have encircled it in red to distinguish it, while maintaining the coloring to show that it's closely related to <i>M</i> . <i>indolicus</i>
FIGURE S1= Are these sequences included in the phylogenetic tree of Figure 2 as well? I would recommend to include these sequences into the Figure 2 for better view of these clusters.	Yes, Figure 2 is a combination of BVAB-2 and BVAB-3 (and associated sequences), but Fig. S1 is only BVAB-3.
RESULTS AND DISCUSSION= Please, report in this section the results and their interpretations only; it is recommended not to mention further facts, such as the current methods used for the BVAB diagnosis. I would recommend to focus the attention more on the potential novel identified species, giving further explanation of the obtained results.	Thanks for the recommendation. Indeed, we only mentioned the diagnosis aspect in the concluding paragraph (line 125) as a supporting argument for the need to know the species of these unknown organisms. The mention of PCR & sequencing was therefore not the focus, but an example of the importance of this work.

June 4, 2023

Dr. John Osei Sekyere University of Pretoria Department of Dermatology School of Medicine Faculty of Health Sciences Pretoria, Gauteng 0084 South Africa

Re: Spectrum04676-22R1 (Species-specific analysis of bacterial vaginosis (BV)-associated bacteria (BVAV))

Dear Dr. John Osei Sekyere:

Your manuscript has been accepted, and I am forwarding it to the ASM Journals Department for publication. You will be notified when your proofs are ready to be viewed.

The ASM Journals program strives for constant improvement in our submission and publication process. Please tell us how we can improve your experience by taking this quick <u>Author Survey</u>.

Publication Fees: We have partnered with Copyright Clearance Center to collect author charges. You will soon receive a message from no-reply@copyright.com with further instructions. For questions related to paying charges through RightsLink, please contact Copyright Clearance Center by email at ASM_Support@copyright.com or toll free at +1.877.622.5543. Hours of operation: 24 hours per day, 7 days per week. Copyright Clearance Center makes every attempt to respond to all emails within 24 hours. For a complete list of **Publication Fees**, including supplemental material costs, please visit our <u>website</u>.

ASM policy requires that data be available to the public upon online posting of the article, so please verify all links to sequence records, if present, and make sure that each number retrieves the full record of the data. If a new accession number is not linked or a link is broken, provide production staff with the correct URL for the record. If the accession numbers for new data are not publicly accessible before the expected online posting of the article, publication of your article may be delayed; please contact the ASM production staff immediately with the expected release date.

Corresponding authors may join or renew ASM membership to obtain discounts on publication fees. Need to upgrade your membership level? Please contact Customer Service at Service@asmusa.org.

Thank you for submitting your paper to Spectrum.

Sincerely,

Steven Frese Editor, Microbiology Spectrum

Journals Department American Society for Microbiology 1752 N St., NW Washington, DC 20036 E-mail: spectrum@asmusa.org